Author's response to reviews

Title: Noncompliance in head and neck cancer treatment; associated factors for both patient and physician

Authors:

Emilie A. C. Dronkers (e.dronkers@erasmusmc.nl)
Steven W. Mes (messteven@gmail.com)
Marjan H. Wieringa (m.vandenbrink@erasmusmc.nl)
Marc P. van der Schroeff (m.vanderschroeff@erasmusmc.nl)
Robert Jan Baatenburg de Jong (r.j.baatenburgdejong@erasmusmc.nl)

Version: 2 Date: 19 February 2015

Author's response to reviews:

Dear sir/madam,

Thank you for your email regarding the editorial changes that are needed in our manuscript.

Hereby we present you our editorially revised manuscript. The following changes are made:

- line numbering is included in the main text, excluding title page and tables
- manually inserted page breaks are removed
- e-mail addresses of each co-author are included on the title page
- the introduction header is renamed as 'background'
- the requested consent statement is included in the Methods section
- a list of abbreviations is provided
- page-layout of tables is changed to 'landscape' for a better view

Kind regards,

On behalf of the co-authors,

Emilie A.C. Dronkers, MD, corresponding author